ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Javdani, Mohsen; McGee, Andrew

Article

Labor market mobility and the early-career outcomes of immigrant men

IZA Journal of Development and Migration

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Javdani, Mohsen; McGee, Andrew (2018) : Labor market mobility and the earlycareer outcomes of immigrant men, IZA Journal of Development and Migration, ISSN 2520-1786, Springer, Heidelberg, Vol. 8, Iss. 20, pp. 1-28, https://doi.org/10.1186/s40176-018-0128-4

This Version is available at: https://hdl.handle.net/10419/197479

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

ORIGINAL ARTICLE

Open Access

Labor market mobility and the early-career outcomes of immigrant men

Mohsen Javdani^{1*} and Andrew McGee^{2,3}

* Correspondence: mohsen. javdani@ubc.ca

¹Department of Economics, 3333 University Way, Kelowna, BC V1V 1V7, Canada Full list of author information is available at the end of the article

Abstract

We examine the role of between- and within-firm mobility in the early-career outcomes of immigrant men. Among Canadian workers with less than 10 years of potential experience, we find that visible minority immigrants were significantly less likely to have been promoted with their initial employers than similar white natives but were just as likely to have moved to new employers over the course of a year between interviews. White immigrants, on the other hand, were just as likely to be promoted as white natives but much more likely to move to new employers—suggesting that they enjoyed more overall mobility than white natives and other immigrants. We present tentative evidence linking these mobility patterns to differences in wage growth and occupational change between immigrants and natives. Overall, our findings suggest that the between- and within-firm mobility of white immigrants may play an important role in their relative economic success in Canada, while adding to growing evidence that visible minority immigrants experience frictions in the labor market that hinder their mobility and thus their economic prospects.

JEL Classification: J61, J71

Keywords: Promotions, Employer changes, Immigration, Visible minorities

1 Introduction

Integrating immigrants into the labor market is a key policy objective in countries with large and growing immigrant populations like Canada, which admitted over five million immigrants between 1995 and 2010 (Statistics Canada 2016) and where 20.6% of residents were born abroad (Chui 2013). Given that many immigrants experience a decline in occupational status when immigrating, studies such as Green (1999), Chiswick et al. (2005), and Abramitzky et al. (2014) have highlighted the importance for immigrants of (upward) occupational mobility.¹ In Canada, the first years following immigration have been shown to be particularly important for immigrants in moving to jobs in their preferred occupations (Grenier and Xue 2011).

In this study, we note that occupational mobility requires job mobility and investigate native-immigrant differences in job mobility early in the career through moves to new employers and within firms via promotions. We then assess the extent to which differences in within- and between-firm mobility contribute to differences in wage growth and occupational mobility between early-career immigrants and natives. To the best of our knowledge, ours is the first study of nativeimmigrant differences in both within- and between-firm mobility.² Furthermore,



© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made.

while an extensive literature documents gender and race differences in promotion outcomes, we provide the first evidence of native-immigrant differences in the rates of and wage returns to promotions.

Our focus on the within- and between-firm mobility of early-career immigrants is further motivated by three important facts from the labor literature. First, there is evidence that immigrants in Canada may encounter search frictions that limit their between-firm mobility. Aydemir and Skuterud (2008) find that the native-immigrant pay gap in Canada can be largely explained by differences in employers as natives tend to be employed in higher wage firms. This suggests that immigrants may face barriers to mobility keeping them in jobs with low-paying firms.³ Along these lines, Oreopoulos (2011) in a resume field experiment finds that the resumes of skilled immigrants in Canada were less likely to elicit contacts from potential employers than resumes from similar natives, while Bowlus et al. (2016) provide evidence of frictions faced by immigrants to Canada in a structural model of job search.

Second, search and matching models imply that all workers (natives and immigrants) benefit on average from early-career moves as they find better matches to their skills (Burdett 1978; Jovanovic 1979)—which is likely even more true for immigrants looking to upgrade their occupations to match their skills. Topel and Ward (1992) found that the average man in the USA has seven jobs in the first 10 years in the labor market and that these movements between jobs account for at least a third of early-career wage growth.⁴ Thus, differences in mobility early in the career may translate into significant differences in earnings over time.

Finally, our interest in within-firm mobility via promotions follows from the observation that workers need not change employers to find better job matches, and an important avenue for changing jobs with an employer is through promotions—which also tend to be concentrated early in the career (Javdani and McGee 2017). Furthermore, promotions are important drivers of wage growth. Among many others, Pergamit and Veum (1999), Cobb-Clark (2001), Francesconi (2001), Blau and DeVaro (2007), and Kosteas (2009) find that promotions are associated with, on average, 5 to 12% increases in wage growth, while McCue (1996) finds that promotions account for 9% of total wage growth in the first 10 years of the career for white men. As a consequence, the failure of immigrants to keep pace with natives in climbing the corporate ladder via promotions early in the career may contribute to native-immigrant wage gaps.

Using a sample of male workers in Canada from 1999 to 2004 from the Workplace and Employee Survey (WES), we estimate the probabilities that natives and immigrants make different transitions over the course of a year between interviews. From one interview to the next, workers enter unemployment, remain with their initial employers without being promoted, remain with their initial employers having been promoted, or move to new employers. We find that visible minority immigrants are 15 percentage points less likely to have been promoted with their current employers but are just as likely to have moved to new employers as white natives. White immigrants, on the other hand, were just as likely to have been promoted with their initial employers as white natives but were as much as 12 percentage points more likely than white natives to move to new employers—suggesting that white immigrants are more mobile than both white natives and visible minority immigrants.

We then examine whether the greater between-firm mobility of white immigrants and lower within-firm mobility of visible minority immigrants translate into differences in wage growth relative to white natives. We note, however, that our analysis of the contributions of mobility to wage growth gaps is necessarily speculative given that the gaps themselves are not precisely estimated. In our sample, the wages of white (visible minority) immigrants grow by 9.5 (6.6) percent on average between interviews compared to 8.3% among white natives. Mobility is clearly related to wage growth in our sample as moves to new employers and promotions are associated with wage growth between interviews of 15 and 2.5%, respectively. Using Oaxaca-Blinder decompositions of the native-immigrant wage growth gaps, we find that—while imprecisely estimated differences between white immigrants and natives in the rates of moves to new employers and promotions can account for a difference in wage growth equal to the whole wage growth gap-driven by white immigrants' higher likelihood of moving to new emplovers. By contrast, visible minority immigrants' slightly higher inter-firm mobility and the large wage gains to employer changes offset the negative effect on wage growth of their lower promotion rates, which renders the total contribution of mobility to the wage growth gap between white natives and visible minority immigrants close to zero.

Finally, we examine the relationship between within- and between-firm mobility and occupational mobility early in the career in light of evidence that immigrants to Canada experience occupational "downgrading" upon arrival.⁵ Here too, the importance of between-firm mobility for white immigrants is evident. Some 91% of white immigrants changing employers switch occupations compared to only 71% of white natives and 81% of visible minority immigrants. As a consequence of their greater likelihood of changing employers and changing occupations conditional on changing employers, 26% of white immigrants in our sample change occupations between interviews compared to only 16% of white natives and 19% of visible minority immigrants. Both moves to new employers and promotions are important channels for changing occupations early in the career in our sample, and white immigrants are particularly good at using the former channel to change—and likely upgrade—occupations.

While Aydemir and Skuterud (2008), Oreopoulos (2011), and Bowlus et al. (2016) provide evidence consistent with visible minority immigrants in Canada encountering job search frictions not encountered by white Canadian-born workers, our primary contribution is to provide direct evidence of actual differences between natives and immigrants in mobility. These differences are surprising because the Canadian immigration system imposed few mobility constraints. Newly arrived permanent residents were not for the most part tied to employers or regions. Likewise, immigrants in Canada on "open work permits" could change employers without restriction, while immigrants on employer-specific work permits needed only apply for a new work permit to change employers. Moreover, work permits could be renewed indefinitely as long as the worker remained employed.

Our second contribution is to note the marked differences among immigrants in mobility patterns and the potential importance of these differences to the diverging fortunes of different groups of immigrants. Specifically, ours is the first study to highlight the role of between-firm mobility in the relative economic success of white immigrants to Canada. This is particularly important for interpreting the existing evidence on mobility among immigrants to Canada. Notably, Skuterud and Su (2012) provide evidence that immigrants to Canada were less likely than natives to transition into high-wage jobs (and more likely to transition out of these jobs), but equally likely to transition into low-wage jobs. Our findings suggest that these patterns may be driven by visible minority (and not white) immigrants.

In terms of interpreting these mobility patterns, we note that the fact that visible minority immigrants were just as likely to move to new employers as white natives should not be taken as evidence against the existence of search frictions. If visible minority immigrants are not promoted at rates commensurate with their skills, presumably they should be pursuing outside options more than white natives. Likewise, immigrants may have stronger incentives to move to new employers in order to upgrade occupations than natives. Both possibilities suggest that visible minority immigrants should be moving to new employers with greater frequency than white natives as we observe among white immigrants. Indeed, Oreopoulos' findings imply that even if visible minority immigrants were sending out resumes at a rate similar to white immigrants, they would generate fewer contacts with employers and thus lead to a lower rate of transitioning to new employers.

We consider three potential explanations for the mobility patterns that we observe: unobserved productivity differences, taste-based discrimination on the part of employers, and information asymmetries and other search frictions. Visible minority immigrants may be less likely to be promoted if they are less productive than natives in ways unobserved by the econometrician. For instance, immigrants may have language difficulties that limit their prospects for promotion. Splitting our sample by age-atimmigration, however, we find that immigrants who arrived in Canada as children actually fare the worse where promotions are concerned. Alternatively, unobserved differences *among* visible minority immigrants in our sample may have resulted from changes in Canadian immigration policy in the early 1990s that prioritized admitting skilled immigrants as opposed to immigrants with family ties. While we find that visible minority immigrants (most of whom arrived as children) fare worse in terms of promotion probabilities than those who arrived after this policy change, the difference in promotion probabilities is not statistically significant.

The lower promotion rates among visible minority immigrants could also arise if employers prefer to promote white workers (natives and immigrants) rather than visible minorities. If this were the case, however, then high-ability visible minority immigrants ought to be more likely to move to new employers as competition gives firms incentives to hire away talented visible minorities experiencing discrimination in promotion outcomes. As a consequence, visible minority immigrants ought, on average, to be more likely to change employers than white immigrants and natives. In our sample, however, visible minority immigrants move to new employers at a similar rate as white natives. Nevertheless, the struggles of visible minority natives observed in our sample in terms of wage growth suggest that taste-based discrimination may play a role in the outcomes of visible minorities regardless of whether they are immigrants.

Of course, taste-based discrimination could persist without visible minority immigrants being more likely to change employers if search frictions exist that prevent visible minority immigrants from moving to new employers. For instance, visible minorities may lack networks for job search or be less familiar with how job search works in Canada. Alternatively, potential employers may have less information about visible minority immigrants. In the "invisibility hypothesis" of Milgrom and Oster (1987), this information asymmetry between current and prospective employers gives employers incentives to "hide" employees about whom the market has less information by denying them promotions that are assumed to convey positive information about the worker to other employers. We find some evidence consistent with such information asymmetries. Specifically, we find that visible minority immigrants about whom the market likely has more information—those with a bachelor's degree or higher—enjoy similar mobility between- and within-firms and wage returns to this mobility as white natives. Ultimately, however, establishing whether information asymmetries or other search frictions lead to the mobility differences between visible minority immigrants and white immigrants and natives requires further investigation, an issue we discuss in the conclusion.

The remainder of the paper proceeds as follows. Section 2 discusses the data as well as the implications of our sample selection criteria. Section 3 discusses the Canadian immigration policies that affected immigrants in our sample and their implications for mobility. Section 4 presents our main findings. Section 5 concludes and poses the questions for future research.

2 Data

Our sample is drawn from the Workplace and Employee Survey (WES), a longitudinal survey of employers and their employees collected by Statistics Canada between 1999 and 2006. In every year, a representative sample of approximately 6000 employers was surveyed.⁶ A maximum of 24 employees were interviewed from each sampled firm in each odd year and re-interviewed the following year regardless of whether they remained with their initial employer.⁷ The employee sample is representative of the Canadian workforce in the target population of employers when properly weighted, and all of our analysis incorporates sample weights from Statistics Canada. While a longer longitudinal dimension would have been preferred, the WES is particularly well-suited for our study insofar promotions and moves to new employers between interviews are well-measured.

Three dependent variables are used in our study. First, we use a categorical variable that identifies the transition made by each worker between interviews to study nativeimmigrant differences in within- and between-firm mobility. A worker either transitions to unemployment (i.e., the employee has left the initial employer and does not have a new employer—including self-employment), transitions to a new employer, remains with the initial employer and has been promoted since the first interview, or remains with the initial employer without having been promoted. Changes in pay and responsibilities are thought to be the distinguishing features of promotions (Pergamit and Veum 1999), and our data identify promotions using precisely these two features. Specifically, whether the employee has been promoted between interviews is based on the questions: "Have you ever been promoted while working for this employer? (By promotion we mean a change in duties/responsibilities that lead to both an increase in pay and the complexity or responsibility of the job)" and "When did your most recent promotion occur?"⁸ The caveat that a promotion must entail a change in job complexity or responsibility is important insofar our interest in promotions stems largely from their role in enabling workers to change occupations. Second, we use the change in the worker's log-hourly wage between interviews to examine the extent to which differences in mobility contribute to differences in wage growth. Third, we create an indicator that equals one when a worker changes occupations between interviews to study the relationship between occupational mobility and between- and within-firm mobility.⁹

Our main analysis is based on the pooled 1999, 2001, and 2003 cross sections of employees; the 2005 cross section cannot be used because WES did not field an employee survey in 2006. We restrict our sample to non-aboriginal men who were interviewed twice with less than 10 years of potential labor market experience (defined as age minus years of schooling minus six).¹⁰ The full sample used to study labor market transitions includes observations from 4907 men after the sample restrictions are imposed—including 260 (266) white (visible minority) immigrants. When studying wage growth and occupation switching, we further restrict the sample to workers who are employed at both interviews (i.e., dropping men who transition to unemployment) resulting in a sample with observations from 4585 men.

We focus on early-career workers for three reasons. First, most job shopping occurs early in the career (e.g., Topel and Ward 1992, van der Klaauw and Dias da Silva 2011). Likewise, most promotion activity occurs early in the career among Canadian workers in the WES (Javdani and McGee 2017). Second, focusing on workers who enter Canada before or shortly after entering the workforce allows us to abstract from issues arising from differences in the returns to labor market experience acquired in different countries that complicate native-immigrant comparisons among older workers. Third, focusing on workers beginning their careers over a single decade enables us to abstract to some extent from differences in macroeconomic conditions upon labor market entry that have affected long-run immigrant and native career trajectories in other cohorts (Green and Worswick 2012).

We create indicators for being a Canadian-born visible minority, a white immigrant, or a visible minority immigrant with white Canadian-born workers serving as the reference category because significant differences exist among both natives and immigrants of different races.¹¹ Our controls include the highest level of schooling, the number of dependent children, an indicator for marital status, a quadratic in age, a quadratic in years of (actual) full-time labor market experience, a quadratic in years of seniority with the current employer, an indicator for full-time employment, an indicator for membership in a union or collective bargaining agreement, an indicator for the language spoken at work being different from the language spoken at home, occupation (six categories), industry (14 categories), and the worker's place in the firm-level wage distribution.¹² We control for the worker's standing in the firm-level wage distribution as a proxy for the worker's position in the firm's hierarchy because different hierarchical levels could be associated with different rates of transitions and the returns to these transitions.

Before discussing the summary statistics, some discussion of our sample selection rules is warranted. Workers who are not interviewed a second time ("attriters") are eliminated from our analysis because we do not observe their employment transitions, wage growth, or occupation switching between interviews. Workers may not be reinterviewed for the usual reasons (e.g., refusals, inability to locate), but immigrants may also attrit because they return to their home country (i.e., population attrition). Systematic, unobserved differences between natives and immigrants may bias our estimates if return migration is correlated with the unobserved attributes of immigrants. To assess the potential importance of non-random attrition, Appendix: Table 9 reports the estimated marginal effects from a probit model of the probability of attrition for visible minority Canadian-born workers, white immigrants, and visible minority immigrants observed in the first (odd year) interview using different sets of controls. While both white and visible minority immigrant men are more likely to attrit than white natives, the difference is only statistically significant for white immigrants.¹³ If this attrition is due to population attrition, our findings should be interpreted as applying to the population of white immigrants who remain in Canada—presumably the population of interest in the long run.¹⁴

Alternatively, it may be the case that workers who change employers between interviews are harder to locate than workers who remain with their initial employers. Attrition along these lines would imply that our estimates understate the between-firm mobility of white immigrants. We have two reasons, however, to doubt that "movers" were more likely to attrit than "stayers." First, the WES documentation indicates following workers who changed employers between interviews was one of the objectives of the survey (Krebs et al. 1999). Second, workers who consented to be interviewed in the odd year submitted forms with their contact information. After 2000, all interviews were done over the phone. The initial employer played no role in contacting workers for the second interview (Krebs et al. 1999). Nevertheless, we acknowledge that there is some potential that our estimates understate the between-firm mobility of immigrants.

Figure 1 details the proportions of each group making particular transitions. Most strikingly, only 20% of visible minority immigrants in our sample were promoted with



their initial employers between interviews relative to 34% of white, Canadian-born men—a difference that is statistically significant at the 1% level.¹⁵ Visible minority immigrants were not significantly more likely to move to new employers relative to white natives (15 versus 12%), but 58% of visible minority immigrants simply remained with their initial employers without being promoted relative to only 46% of white natives—again a statistically significant difference. By contrast, white immigrants were nearly as likely as white natives to be promoted when remaining with their initial employer but significantly more likely to move to new employers.

Table 1 reports summary statistics for each group. In addition to the differences in mobility observed in Fig. 1, immigrants and natives differ in both their wage growth and the rates at which they switched occupations. Early-career white natives experienced wage growth of 8.3% between interviews relative to only 6.6% for visible minority immigrants. White immigrants, by contrast, experienced 9.5% wage growth between interviews.¹⁶ While not statistically significant at conventional levels, the economic significance of these wage growth gaps early in the career could be considerable. Finally, more than a quarter of white immigrants changed occupations between interviews relative to only 16% of white natives.

Table 1 also makes it clear that controlling for observed characteristics may be important as immigrants and natives differ significantly on several dimensions. Consistent with Canada's bias in favor of skilled immigrants discussed in the next section, 51% of visible minority immigrants and 44% of white immigrants in our sample had a bachelor's degree or higher compared to only 19% of white natives. Given that the sample restriction is based on potential experience and immigrants spend more years in school, both visible minority and white immigrants were also on average approximately 2 years older than natives in our sample.

Finally, immigrants were distributed very differently across industries and occupations than their white native peers. For instance, nearly 32% of visible minority immigrants in our sample worked in finance and insurance or business services compared to only 15% of white natives. Similarly, more than 40% of white and visible minority immigrants worked either as managers or as professionals while less than 38% worked in technical occupations or the trades. By contrast, only 25% of white natives worked as managers or professionals while 50% worked in technical occupations and the trades. In Section 4, we examine whether these observed differences between natives and immigrants can explain the unconditional native-immigrant differences in mobility, wage growth, and occupation switching.

3 Immigration policy in Canada

Immigrants in our sample arrived in Canada between 1966 and 2002. In this section, we briefly discuss the key features of and changes to Canadian immigration policy in this period and the likely implications for immigrants' mobility. In 1967, a points system to score applicants based on characteristics such as education, age, language, and occupation was introduced to provide an objective standard for admission to Canada. Three main admission classes were established: economic-class applicants whose eligibility was evaluated solely based on the point system, nominated relatives who were assessed under the point system but received bonus points based on kinship, and family class applicants who were admitted solely on family ties.

Table 1 Summary statistics

Variable	White Canadian-born	Visible minority Canadian-born	White immigrant	Visible minority immigrant
Unemployed	0.082	0.084	0.043	0.067
New employer	0.123	0.110	0.209	0.151
Same employer, promoted	0.335	0.269	0.311	0.199
Same employer, not promoted	0.460	0.537	0.437	0.583
Change in log-hourly wage [‡]	0.083	0.009	0.095	0.066
Changed occupations [‡]	0.162	0.223	0.258	0.193
Log-hourly wage in the first interview	2.664	2.773	2.887	2.731
Education				
Bachelor or higher	0.190	0.305	0.444	0.509
More than high school but less than bachelor	0.551	0.548	0.356	0.389
High school grad	0.181	0.074	0.141	0.077
Less than high school*	0.077	0.071	0.056	0.023
Married	0.426	0.243	0.547	0.437
Number of dependent children	0.282	0.109	0.272	0.298
Age	25.202	24.925	27.397	27.596
Years of actual full-time experience	5.281	4.427	5.735	5.481
Full-time job	0.737	0.615	0.822	0.862
Membership in a union	0.181	0.161	0.109	0.144
Tenure with employer	3.044	2.201	3.199	2.833
Language spoken at home different from work	0.032	0.020	0.014	0.053
Industry				
Construction and natural resources	0.085	0.015	0.063	0.007
Transportation, warehousing, wholesale	0.116	0.044	0.128	0.065
Communication and other utilities	0.017	0.014	0.007	0.011
Retail trade and consumer services	0.300	0.238	0.222	0.263
Finance and insurance	0.027	0.123	0.029	0.120
Real estate, rental, and leasing operations	0.016	0.019	0.007	0.004
Business services	0.124	0.144	0.175	0.197
Education and health services	0.071	0.163	0.075	0.057
Information and cultural industries	0.037	0.047	0.038	0.021
Manufacturing*	0.202	0.188	0.250	0.250
Occupation				
Managers	0.093	0.074	0.150	0.150
Professionals	0.142	0.261	0.323	0.251
Technical/traders	0.504	0.332	0.374	0.354
Marketing/sales	0.054	0.141	0.019	0.071
Clerical/administrative	0.087	0.151	0.052	0.121
Production workers*	0.116	0.038	0.079	0.050
Percentile in firm wage distribution	0.298	0.321	0.256	0.341
Number of observations	4192	189	260	266

*Omitted category in our regression specifications *The statistic applies to the restricted sample of 4585 men who are employed in both interviews

In 1978, a new Immigration Act prioritized the admission of family members and refugees—thereby reducing the share of immigrants admitted under the economic class, who already constituted a small share of admitted applicants. Further changes in 1982 limited the admission of economic class applicants to those with pre-arranged employment, but these restrictions proved to be short-lived. Concerns about Canada's low fertility rate and an aging population in 1986 resulted in the elimination of the prearranged employment requirement for economic class applicants and a substantial increase in immigration levels with the number of immigrants admitted annually rising from 83,000 in 1985 to 99,000 in 1986 and ultimately to 250,000 by 1993 (Green and Green 1999).

In the early 1990s, Canada's immigration policy moved from emphasizing family reunification and short-term occupational needs to an emphasis on growing the country's stock of human capital. To this end, the share of family class immigrants was reduced in favor of economic class immigrants even as annual inflows of immigrants remained stable at about 1% of the population (Green and Green 1999). As a consequence, the composition of immigrants to Canada changed substantially in this period with significant increases in the average education level of newly arrived immigrants and the number of visible minority immigrants. This shift in policy has implications for the immigrants in our sample. Immigrants who entered Canada in the 1990s and early 2000s either as dependent children or applicants would likely have been selected based on their (or their parents') skill levels, while immigrants who entered Canada as young children prior to the 1990s would not necessarily come from families with high skill levels.¹⁷ The effect on mobility of the change in immigration regimes, however, is unclear as child immigrants in our sample who entered under the former policy would also benefit from greater language acquisition and cultural assimilation.

In the late 1990s, Provincial Nominee Programs (PNP) were introduced that allowed provincial governments to nominate applicants for immigration based on the provinces' labor market needs; the federal government remains responsible for admitting nominees. Most of the PNPs—which differ by province in their particulars—require an applicant to work in the nominating province for a period of time on a temporary work permit before applying, but immigrants are not tied to a specific employer provided they remain in the province.¹⁸ As such, immigrants entering Canada through the PNPs might be expected to be less mobile between firms than other workers as the universe of potential employers is restricted. These PNP restrictions on mobility, however, were unlikely to have affected many immigrants entering Canada in our sample period was trivial. In 1999, for instance, only 477 immigrants entered Canada under the PNPs, and less than 3% (6248) of immigrants in 2004 were admitted via PNPs (Citizenship and Immigration Canada 2011).

To summarize, no immigration policies in place during our sample period restricted the mobility of immigrants between employers within a given province. In the period in which immigrants in our sample entered Canada, however, immigration selection procedures changed significantly, and the composition of the immigrant population changed significantly as well. In the next section, we examine whether the change in immigration selection procedures led to unobserved changes in immigrants that affected immigrants' mobility.

4 Findings

4.1 Immigration and between- and within-firm mobility

We first estimate multinomial logit models of the probabilities of making each transition between interviews. Each panel of Table 2 concerns a single transition. The first row of each panel reports the predicted probability of the transition for white, Canadian-born men. Below this predicted probability, each row reports the estimated difference between the predicted probabilities of making the transition for the specified minority group and white natives. Column (1) of Table 2 reports the estimates including only indicators for group membership. Columns (2) to (6) add controls for worker and job characteristics, occupation, industry, and the worker's position in the firm's wage distribution.

A worker's occupation and industry may be endogenous if workers select into particular industries and occupations based on unobserved characteristics related to their employment transitions. Likewise, the worker's standing in the employer's wage distribution may be endogenous insofar it likely reflects earlier transitions. *None of the controls*, however, appreciably affect the estimated differences in the probabilities of each transition between white native men and the members of each minority group.

Conditional on worker and job characteristics in column (2), the estimated probability of transitioning to unemployment between interviews for white native men in their first 10 years in the labor market is 0.056. The probabilities of transitioning to unemployment for members of every other group are statistically indistinguishable from and within 1 percentage point of the estimated probability for white natives in each specification in columns (2) to (6). This is particularly important insofar we restrict our sample to workers who remain employed in the second interview in subsequent analysis relating transitions to wage growth. The similar estimated probabilities of transitioning to unemployment across groups suggests that this restriction is unlikely to materially affect our inferences.

The second panel of Table 2 reports the predicted probabilities of remaining with the initial employer and being promoted. Similar to the unconditional difference in Fig. 1, visible minority immigrants are approximately 15 percentage points less likely to remain with their initial employer having been promoted than white natives in columns (2) to (6). While we also find that visible minority natives and white immigrants are less likely to have been promoted than similar white natives, these differences are never statistically distinguishable from zero.

In the third and fourth panels, we find that visible minority immigrants are approximately 8 percentage points more likely to remain with the initial employer without being promoted and 6 percentage points more likely to move to a new employer between interviews than similar white natives—although neither difference is statistically significant.¹⁹ By contrast, white immigrants are nearly 12 percentage points more likely to move to new employers between interviews than white natives—a difference that is statistically distinguishable from zero at the 10% level in most specifications and highlights the differences in mobility among immigrants with different ethnic backgrounds. While white immigrants exhibit a higher degree of interfirm mobility than white natives, they enjoy a similar probability of promotion when remaining with their initial employers. Visible minority immigrants, on the other hand, are also slightly more likely than white natives to move to new employers but are much less likely than white natives to be promoted when remaining with their initial employers. Differences between the groups in observed characteristics cannot explain the differences in early-career mobility evident in Fig. 1.²⁰

Table 2 Multinomial logit estimates of transition probabilities

	(1)	(2)	(3)	(4)	(5)	(6)
Outcome: unemployment						
White Canadian-born	0.075***	0.056***	0.055***	0.053***	0.053***	0.053***
(predicted probability)	(0.009)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)
Visible minority Canadian-born	0.006	- 0.010	- 0.010	- 0.004	- 0.005	- 0.008
	(0.032)	(0.020)	(0.019)	(0.020)	(0.020)	(0.020)
White immigrant	- 0.036**	- 0.010	- 0.006	- 0.009	- 0.005	- 0.005
	(0.017)	(0.018)	(0.019)	(0.018)	(0.019)	(0.019)
Visible minority immigrant	- 0.012	0.008	0.008	0.007	0.008	0.006
	(0.024)	(0.025)	(0.025)	(0.024)	(0.024)	(0.024)
Outcome: same employer and pror	moted					
White Canadian-born	0.337***	0.356***	0.357***	0.353***	0.354***	0.356***
(predicted probability)	(0.017)	(0.018)	(0.018)	(0.018)	(0.018)	(0.018)
Visible minority Canadian-born	- 0.063	- 0.060	- 0.066	- 0.054	- 0.057	- 0.055
	(0.070)	(0.072)	(0.069)	(0.071)	(0.069)	(0.069)
White immigrant	- 0.025	- 0.040	- 0.049	- 0.027	- 0.036	- 0.029
	(0.051)	(0.055)	(0.055)	(0.055)	(0.055)	(0.055)
Visible minority immigrant	- 0.135***	- 0.148***	- 0.147***	- 0.151***	- 0.150***	- 0.153***
	(0.042)	(0.044)	(0.044)	(0.043)	(0.043)	(0.043)
Outcome: same employer and not	promoted					
White Canadian-born	0.462***	0.485***	0.489***	0.491***	0.495***	0.495***
(predicted probability)	(0.017)	(0.018)	(0.018)	(0.017)	(0.017)	(0.017)
Visible minority Canadian-born	0.071	0.093	0.092	0.076	0.072	0.069
	(0.072)	(0.072)	(0.070)	(0.068)	(0.068)	(0.070)
White immigrant	- 0.024	- 0.066	- 0.069	- 0.077	- 0.078	- 0.055
	(0.054)	(0.056)	(0.056)	(0.058)	(0.058)	(0.056)
Visible minority immigrant	0.119**	0.078	0.079	0.081	0.083	0.088
	(0.060)	(0.066)	(0.066)	(0.064)	(0.065)	(0.065)
Outcome: new employer						
White Canadian-born	0.125***	0.102***	0.098***	0.102***	0.098***	0.096***
(predicted probability)	(0.012)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)
Visible minority Canadian-born	- 0.013	- 0.023	- 0.015	- 0.018	- 0.009	- 0.005
	(0.040)	(0.030)	(0.031)	(0.034)	(0.036)	(0.037)
White immigrant	0.085	0.117*	0.125*	0.114*	0.121*	0.089
	(0.063)	(0.064)	(0.066)	(0.066)	(0.068)	(0.062)
Visible minority immigrant	0.027	0.061	0.058	0.061	0.058	0.057
	(0.049)	(0.056)	(0.055)	(0.054)	(0.055)	(0.054)
Controls						
Personal and job characteristics		Yes	Yes	Yes	Yes	Yes
Occupation			Yes		Yes	Yes
Industry				Yes	Yes	Yes
Percentile in wage distribution						Yes

Notes: Robust standard errors clustered at the firm level are reported in parentheses. The estimation sample includes observations from 4907 men. For white Canadians, we report the predicted probability of each transition, while for minority groups, we report the difference between the predicted probabilities for each group and white Canadians *Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

Before examining the role of these mobility differences in native-immigrant differences in wage growth and occupational change, we briefly consider some potential explanations for the native-immigrant differences in mobility observed in Table 2. Immigrating at an earlier age presumably leads to greater language competency, which has been shown to affect native-immigrant wage differentials (e.g., Chiswick and Miller 1995, Dustmann and van Soest 2002, Bleakley and Chin 2004, Adsera and Ferrer 2015).²¹ Researchers have speculated that a "critical age" exists after which perfect language acquisition (i.e., vocabulary, syntax, accent) is impossible (Singleton and Lengyel 1995). Using age 9 as a rough benchmark for the critical age in Canada, we estimate the transition probabilities for immigrants who immigrated before age 9 and those who immigrated after age 9.²²

Table 3 reports these estimates by age-at-immigration.²³ Perhaps surprisingly, visible minority immigrants who arrived in Canada *before age 9* are an estimated 27 percentage points less likely than similar white natives to be promoted with the initial employer between interviews while being 16.8 percentage points more likely to move to new employers. Visible minority immigrants who move to Canada after age 9, on the other hand, are only 9.3 percentage points less likely to have been promoted. Language competency appears unlikely to be the dominant factor underlying the struggles of visible minority immigrants in internal labor markets.

Complicating the interpretation of the estimates in Table 3, however, is the shift in Canadian immigration policy in the early 1990s discussed in the previous section. Most (but not all) of the early-career immigrants in the WES who arrived in Canada before age 9 would have arrived under the older policy placing less emphasis on the skill of their parents, while most (but not all) immigrants who arrived after age 9 would have

Group	(1)	(2)	(3)	(4)
	Unemployment	New employer	Promoted with initial employer	Not promoted with initial employer
White Canadian-born	0.056***	0.102***	0.356***	0.486***
	(0.008)	(0.010)	(0.019)	(0.018)
Visible minority Canadian-born	- 0.010	- 0.023	- 0.061	0.094
	(0.020)	(0.031)	(0.072)	(0.072)
White immigrant entering Canada before age 9	- 0.001	0.120	- 0.037	- 0.083
	(0.027)	(0.117)	(0.086)	(0.094)
White immigrant entering	- 0.020	0.113*	- 0.040	- 0.053
Canada at age 9 or older	(0.022)	(0.068)	(0.069)	(0.065)
Visible minority immigrant entering Canada before age 9	- 0.028	0.168	- 0.272***	0.132
	(0.019)	(0.142)	(0.041)	(0.137)
Visible minority immigrant	0.033	0.008	- 0.093*	0.051
entering Canada at age 9 or older	(0.039)	(0.038)	(0.056)	(0.075)

Table 3 Transition probabilities by age-at-immigration

Notes: Robust standard errors clustered at the firm level are reported in parentheses. The estimation sample includes observations from 4907 men. The table reports estimates from multinomial logit models of transition probabilities similar to those in column (2) of Table 2 controlling for worker and job characteristics. We replace the visible minority and white immigrant indicators in Table 2 with four immigrant indicators (visible minority or white immigrants entering Canada before age 9 or at 9 years of age or older). Slightly more than a third of immigrants in our sample enter Canada before age 9. For white Canadians, we report the predicted probability of each transition, while for visible minority immigrants, we report the difference in the predicted probability between visible minority immigrants in a particular age-at-immigration group and white Canadian-born workers

*Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

been admitted after Canada began emphasizing immigrants' skill. If unobserved skill levels are correlated within families, the immigrants in our sample who arrived early in life might be less skilled in unobserved senses than other immigrants. In Table 4, we allow the transition probabilities for immigrants to depend on whether immigrants entered Canada before or after 1993.²⁴ The estimates suggest that visible minority immigrants who entered Canada before 1993 and after 1993 were 16.2 and 12.5 percentage points less likely than white natives to have been promoted between interviews, respectively—a difference that is not statistically significant. This suggests that the change in immigrants observed in Table 2.

Alternatively, the struggles of visible minority immigrants in internal labor markets may stem from potential employers discounting the signaling value of immigrants' foreign credentials. If so, immigrants with more credentials to be discounted—more educated immigrants—might experience the greatest impediments to between- and withinfirm mobility relative to similar white, Canadian-born men. To test this hypothesis, we report in Table 5 estimates from separate multinomial logit models for workers with and without a bachelor's degree or higher. For visible minority and white immigrants with bachelor's degrees, we fail to reject the null hypotheses that their transition probabilities are identical to those of white natives in column (1). Among workers without a bachelor's degree in column (2), however, visible minority immigrants are an estimated 19.4 percentage points less likely to be promoted while remaining with the initial employer without being significantly more likely to move to new employers than white natives. Insofar internal mobility is concerned, credential discounting does not appear to drive the immigrant-native differences in mobility observed in Table 2.

•		0		
Group	(1)	(2)	(3)	(4)
	Unemployment	New employer	Promoted with initial employer	Not promoted with initial employer
White Canadian-born	0.056***	0.102***	0.356***	0.485***
	(0.008)	(0.010)	(0.019)	(0.018)
Visible minority	- 0.010	- 0.023	- 0.060	0.093
Canadian-born	(0.020)	(0.031)	(0.072)	(0.072)
White immigrant entering	- 0.004	0.151*	- 0.072	- 0.075
Canada before 1993	(0.023)	(0.084)	(0.064)	(0.069)
White immigrant entering	- 0.029	0.025	0.045	- 0.041
Canada in 1993 or later	(0.019)	(0.059)	(0.088)	(0.089)
Visible minority immigrant	0.008	0.082	- 0.162***	0.073
entering Canada before 1993	(0.032)	(0.084)	(0.055)	(0.091)
Visible minority immigrant	0.010	0.030	- 0.125*	0.084
entering Canada in 1993 or later	(0.041)	(0.055)	(0.070)	(0.094)

Table 4 Transition probabilities by year of immigration

Notes: Robust standard errors clustered at the firm level are reported in parentheses. The estimation sample includes observations from 4907 men. The table reports estimates from multinomial logit models of transition probabilities similar to those in column (2) of Table 2 controlling for worker and job characteristics. We replace the visible minority and white immigrant indicators in Table 2 with four immigrant indicators (visible minority or white immigrants entering Canada before 1993 and in 1993 or later). Around a third of immigrants in our sample enter Canada after 1993. For white Canadians, we report the predicted probability of each transition, while for visible minority immigrants, we report the difference in the predicted probability between visible minority immigrants in a particular age-at-immigration group and white Canadian-born workers

*Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

That less-educated visible minority immigrants in Canada struggle in internal labor markets is consistent with Milgrom and Oster's (1987) "invisibility hypothesis." Milgrom and Oster assume that potential employers possess less information about the ability of disadvantaged workers—rendering such workers "invisible" to potential employers. Promotions are assumed to convey (positive) information about these workers to other employers. Current employers with private information regarding their highability but "invisible" workers have incentives to conceal them by limiting their promotion opportunities. This suppresses the signals of ability promotions send to competing employers and prevents these workers from being bid away by other firms.

If employers have less information about the productivity of visible minority immigrants—particularly less educated ones—and promotions signal ability to asymmetrically informed firms, visible minority immigrants would be less likely to be promoted compared to white natives as we document given employers' incentives to "conceal" these workers. This would be less likely to be true for white immigrants—many of whom come from the USA and other parts of the Commonwealth—about whom employers may have better information.²⁵ Furthermore, the lower probability of promotion for visible minority immigrants would not necessarily lead to a disproportionately higher probability of between-firm mobility for visible minority immigrants because other employers possess less information about high-ability, visible minority immigrants and thus would be less likely to offer wages higher than their current employers.

Alternatively, other search frictions—including taste-based discrimination—may limit the outside opportunities of visible minority immigrants. Employers may decline to promote such workers precisely because the employer does not need to compete with outside offers. We discuss the need for further research on the nature of search frictions experienced by visible minority immigrants in the conclusion.

4.2 Mobility and wage growth

To establish the importance of mobility to wage growth in our sample, panel A of Table 6 reports estimates from log-wage growth models in which we regress the change in log-hourly wages between interviews on indicators for whether workers have been promoted with their initial employers or changed employers between interviews as well as different sets of controls. Men who remain with their initial employers without having been promoted serve as the reference group, and the sample necessarily excludes those workers who transition to unemployment. Moving to new employers between interviews is associated with wage growth of 15.7% in our sample controlling for worker and job characteristics in column (2)—larger than the 10% wage growth associated with job transitions reported by Topel and Ward (1992). The estimated wage growth associated with promotions is 2.5% in column (2)—somewhat smaller than Cobb-Clark's (2001) estimate of 4.5% among early-career men in the NLSY79. The estimates in columns (2) to (6) indicate that these returns are not sensitive to the choice of controls.

In panel B of Table 6, we report estimates in which we allow the returns to transitions to differ by group for the specification controlling for worker and job characteristics. The p values for Wald tests of the hypotheses that the returns to a given transition are the same across groups are given in column (5). For transitions to new employers, we fail to reject the null of equal returns across groups. For promotions, however, the p

	With a bachelor's degree or higher degree	Without a bachelor's degree
	(1)	(2)
Outcome: unemployment		
White Canadian-born	0.030***	0.065***
(predicted probability)	(0.009)	(0.009)
Visible minority Canadian-born	- 0.025**	- 0.000
	(0.010)	(0.030)
White immigrant	- 0.014	- 0.005
	(0.014)	(0.027)
Visible minority immigrant	0.004	0.016
	(0.030)	(0.035)
Outcome: different employer		
White Canadian-born	0.050***	0.115***
(predicted probability) Visible minority Canadian-born	(0.016)	(0.011)
Visible minority Canadian-born	- 0.039**	- 0.006
	(0.017)	(0.044)
White immigrant	0.069	0.093
	(0.065)	(0.069)
Visible minority immigrant	0.023	0.069
	(0.043)	(0.072)
Outcome: same employer and promoted		
White Canadian-born (predicted probability)	0.342***	0.356***
	(0.029)	(0.021)
Visible minority Canadian-born	- 0.012	- 0.084
	(0.109)	(0.093)
White immigrant	0.017	- 0.040
	(0.081)	(0.068)
Visible minority immigrant	- 0.085	- 0.194***
	(0.067)	(0.050)
Outcome: same employer and not promo	oted	
White Canadian-born	0.577***	0.464***
(predicted probability)	(0.030)	(0.020)
Visible minority Canadian-born	0.076	0.092
	(0.110)	(0.089)
White immigrant	- 0.072	- 0.047
	(0.084)	(0.070)
Visible minority immigrant	0.057	0.108
	(0.088)	(0.084)
# of observations	1275	3632

Table 5 Multinomial logit regressions by education

Notes: See the notes for Table 2. We separately estimate the multinomial logit models for workers with a bachelor's degree and those without such a degree and report the estimates in columns (1) and (2) *Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

value (less than 0.01) strongly supports rejecting the null, but this is driven entirely by visible minority natives, who experience much smaller wage growth following promotions than other workers. For white and visible minority immigrants, we fail to reject

	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: wage returns to promo	otion and e	employer a	change same	e for all groups		
1 if promoted	0.024*	0.025*	0.025*	0.027*	0.026*	0.025*
	(0.013)	(0.014)	(0.014)	(0.013)	(0.014)	(0.013)
1 if changed employer	0.150***	0.157***	0.156***	0.160***	0.158***	0.160***
	(0.030)	(0.028)	(0.027)	(0.028)	(0.027)	(0.028)
Controls						
Person and job characteristics		Yes	Yes	Yes	Yes	Yes
Occupation			Yes		Yes	Yes
Industry				Yes	Yes	Yes
Percentile in firm wage distribution						Yes
	(1) White native	(2) Visible minority native	(3) White immigrant	(4) Visible minority immigrant	(5) p value (1) = (2) = (3) = (4)	
Panel B: allowing wage returns	to vary by	group				
1 if promoted	0.027	_ 0.107***	0.021	0.034	0.001	
	(0.017)	(0.035)	(0.027)	(0.026)		
1 if changed employer	0.015***	- 0.013	0.209**	0.146*	0.416	
	(0.032)	(0.134)	(0.106)	(0.086)		

Table 6 Wage returns to promotion and employer change

Notes: Panel A reports the estimated coefficients of indicators for being promoted or changing employers between interviews from regressions of log-wage growth between interviews on these transition indicators and the controls listed for each column. Columns (1) to (4) of panel B report the estimated returns to promotions and employer changes for individuals belonging to different groups. The log-wage growth regression in panel B includes respondent and job characteristics (corresponding to column (2) in panel A) and interactions between the transition indicators and indicators for group membership (with white Canadian-born workers serving as the reference group). Panel B reports the sum of the coefficient estimates for the transition indicators and these interaction terms. The standard errors were calculated using the *lincom* command in Stata. Column (5) reports the *p* value for the *F* test of the null hypothesis that the interactions between the group indicators (for being a visible minority Canadian-born worker, a white immigrant, or a visible minority immigrant) and the transitions indicator (promotion or employer change) given in the far-left column are jointly equal to zero

*Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

the null that the returns to promotion are the same as for white natives. If unobserved productivity differences or taste-based discrimination (in combination with search frictions) were behind the differing experiences of white natives and visible minority immigrants, we might expect the wage returns to these transitions to vary by group, but they do not. Instead, the similar returns to transitions across groups are consistent with the "invisibility hypothesis" discussed above insofar less visible workers—while less likely to be promoted—are expected to receive wage increases following promotions comparable to other workers when they manage to get promoted.

We next examine the link between mobility differences and wage growth across groups. As noted in the introduction, however, this exercise is limited by the fact that the wage growth gaps themselves are not precisely estimated. Panel A in Table 7 details the average log-wage growth experienced by members of each group and the wage growth gaps between white natives and the minority groups in our sample. Wage growth early in the career is quite rapid. White natives enjoyed average wage growth between interviews of 8.3%, while white (visible minority) immigrants experienced wage growth between interviews of 9.5 (6.6) percent. While the gaps in wage growth relative

to white native men are not statistically significant at conventional levels, we note that they would generate large wage gaps in wage levels if compounded over several years.²⁶ Visible minority natives as a group are again an outlier in our sample insofar they experienced little wage growth between interviews.²⁷

In panel B of Table 7, we report estimates of the Oaxaca-Blinder (O-B) decompositions of the gaps in average wage growth $(\widehat{\Delta wg}_M)$ between white natives (WN) and the minority groups (*M*) (Blinder 1973; Oaxaca 1973). For each minority group, the decomposition takes the form

$$\widehat{\varDelta wg}_{M} = \sum_{k=1}^{k} \hat{\beta}_{\mathrm{WN},k} \big(\overline{X}_{\mathrm{WN},k} - \overline{X}_{M,k} \big)_{\boxtimes} + \sum_{k=1}^{k} \Big(\hat{\beta}_{\mathrm{WN},k} - \hat{\beta}_{M,k} \Big) \overline{X}_{M,k}$$
explained gap
unexplained gap

The O-B decomposition estimates the contributions to the log-wage growth gap between white natives and the members of a given minority group of observed differences in transitions and characteristics (referred to as the "explained" gap) and differences between groups in the "returns" to these characteristics (referred to as the "unexplained" gap).²⁸

Consistent with white natives being much more likely to be promoted than visible minority immigrants, the estimates in column (1) indicate that the difference in promotion receipt can account for 0.4 percentage points of the 1.7 percentage point gap in wage growth between white natives and visible minority immigrants—a contribution statistically significant at the 10% level. Oaxaca and Ransom (1999), however, show that only the total effect of the full set of categorical dummies is identified. Bearing this in mind, we note that transitions to new employers contribute – 0.4 percentage points to the observed wage growth gap between white natives and visible minority immigrants because a higher proportion of visible minority immigrants move to new employers than white natives and the wage returns to such moves are very high. As a result, the total contribution of within-firm and between-firm mobility to the wage growth gap between white natives is approximately zero.

The O-B estimates in column (2) indicate that labor market transitions (moves to new employers and promotions) can account for 1.2 percentage points of the 1.1 percentage point gap in wage growth between white natives and white immigrants that favors white immigrants. This is unsurprising given that white immigrants in our sample were much more likely than white natives to change employers and the wage returns to employer changes are very large. While imprecisely estimated, we note that this contribution of mobility to the wage growth gap enjoyed by white immigrants is larger than the contribution of any other observable (e.g., experience, education). Thus, we tentatively infer that between-firm mobility may be important in explaining the early-career success of white immigrants relative to their white native peers.²⁹

4.3 Mobility and occupation switching

Standard models of job search suggest that young workers shop jobs for good matches (Jovanovic 1979). Moreover, the early part of the career likely entails a period of occupational experimentation as young people learn about their own skills and the demands of different occupations (Antonovics and Golan 2012). Young immigrant men may

	Visible minority immigrant	White immigrant	Visible minority Canadian-born
	(1)	(2)	(3)
Log wage growth for white Canadians	0.083***	0.083***	0.083***
	(0.007)	(0.007)	(0.007)
Log wage growth for minority group	0.066**	0.095***	0.009
	(0.026)	(0.027)	(0.030)
Raw gap in log wage growth	0.017	- 0.011	0.073**
	(0.027)	(0.028)	(0.031)
Explained	-0.014	- 0.026*	- 0.004
	(0.010)	(0.013)	(0.011)
Unexplained	0.032	0.014	0.078**
	(0.026)	(0.025)	(0.032)
Explained by differences in characteristics			
Transition: new employer	- 0.004	- 0.013	0.002
	(0.008)	(0.010)	(0.006)
Transition: promoted	0.004*	0.001	0.002
	(0.002)	(0.001)	(0.002)
Demographics	- 0.004	- 0.007	0.007*
	(0.008)	(0.008)	(0.004)
Education	- 0.006	- 0.003	- 0.005
	(0.007)	(0.005)	(0.004)
Experience	- 0.002	- 0.003	0.003
	(0.002)	(0.003)	(0.004)
Occupation	0.004	0.006	0.000
	(0.004)	(0.004)	(0.005)
Industry	- 0.008*	- 0.003	- 0.010
	(0.004)	(0.003)	(0.006)
Explained by differences in returns to charac	cteristics		
Transition: new employer	- 0.004	0.001	0.013
	(0.016)	(0.013)	(0.016)
Transition: promoted	0.002	- 0.002	0.025
	(0.011)	(0.011)	(0.022)
Demographics	- 0.428	- 2.874***	1.146
	(1.335)	(0.870)	(1.779)
Education	0.048	0.010	- 0.020
	(0.040)	(0.020)	(0.040)
Experience	0.086	0.025	0.015
	(0.082)	(0.070)	(0.060)
Occupation	- 0.008	0.004	0.016
	(0.022)	(0.018)	(0.022)
Industry	0.023	- 0.024	- 0.067*
	(0.026)	(0.023)	(0.036)

Table 7 Oaxaca-Blinder decomposition of the wage growth gap between white natives and members of minority groups

Notes: Robust standard errors in parentheses. The sample includes men who do not transition to unemployment *Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

have even more motivation to change occupations if they enter the Canadian labor market in occupations for which they are over-qualified as documented in Wald and Fang (2008).

To assess the relationship between occupational mobility and between- and withinfirm mobility among immigrants and natives, we report in panel A of Table 8 the fraction of each group changing occupations. Between interviews, 16 (22) percent of white (visible minority) natives change occupations. By contrast, nearly 26% of white immigrants and 19% of visible minority immigrants switch occupations.

In panel B, we report the fraction of workers making each transition who change occupations. White immigrants who move to new employers switch occupations 91% of the time. By contrast, only 71% of white natives and 80% of visible minority immigrants and natives switch occupations when moving to new employers. Among workers promoted and not promoted remaining with the initial employer, immigrants and white natives switched occupations at around the same rate.³⁰

Finally, panel C reports the fraction of occupation switchers in each group who made a particular transition. The percentages in this panel reflect both the rates of occupation switching among workers making different transitions reported in panel B and the rates of each transition among workers in different groups reported in Table 1. The means in panel C indicate that immigrants move between occupations in very different ways relative to white natives. Among white natives, 59% of the occupation switchers did so by moving to new employers, while 77 and 68% of white and visible minority immigrants who switched occupations, respectively, did so by moving to new employers. Internal promotions account for a much larger fraction (32%) of occupation switching among white natives. By contrast, only 18 and 19% of occupation switches for white and visible minority immigrants, respectively, were realized through internal promotions. For white immigrants, this is largely because white immigrants were more likely to move to new employers and were more successful at switching occupations in these

	5			
	White natives	Visible minority natives	White immigrants	Visible minority immigrants
	(1)	(2)	(3)	(4)
Fraction sw	itching occup	pations between inter	views	
	0.162	0.223	0.258#	0.193
Fraction switching	g occupations	s when making a give	n transition	
New employer	0.713	0.793	0.910**	0.814
Promoted with initial employer	0.142	0.386 [#]	0.141	0.171
Not promoted with initial employer	0.029	_a	0.029	0.039
Fraction of occupation s	witchers in a	group who made a p	articular transition	
New employer	0.591	_b	0.770	0.684
Promoted with initial employer	0.318	0.506	0.178	0.189
Not promoted with initial employer	0.090	_a	0.052	0.126

Table 8 Transitions and occupation switching

Notes: The indicators for statistical significance refer to the differences between the statistic in each column and that in column (1) for white, Canadian-born men

*Significant at the 15% level; *significant at the 10% level; **significant at the 5% level

^aFraction not released by Statistics Canada because the cell (visible minority natives who remain with the initial employer but were not promoted by who switch occupations between interviews) contains fewer than five individuals

^bFraction not released by Statistics Canada because in fractions in this column must sum to 1, meaning that the fraction of occupation switches accounted for by workers who were not promoted among visible minority natives if this fraction were released

moves than white natives. The difference in the rates of occupation switching between white natives and visible minority immigrants, however, is driven more by the fact that visible minority immigrants are much less likely than white natives to be internally promoted.

5 Discussion and conclusions

Our study presents two important stylized facts about mobility patterns between and within employers among early-career natives and immigrants in Canada. First, visible minority immigrants were much less likely to be promoted with their initial employers than white natives while being similarly likely to change employers between interviews. Second, white immigrants were much more likely than white natives to change employers while being just as likely be promoted with their initial employers. We present tentative evidence linking this greater between-firm mobility of white immigrants to their relatively fast wage growth and their ability to change occupations. Overall, our findings suggest that mobility may play an important role in the relative economic success of early-career white immigrants.

Important questions remain concerning the role of between- and within-firm mobility in the assimilation of immigrants. First, how does mobility influence the experiences of immigrants over a longer horizon? A major limitation of the WES is that the longitudinal component for workers is limited to a single year between interviews. Observing the contributions of mobility to the experiences of immigrants over a longer period, however, may be important. Both Topel and Ward (1992) and Light and McGarry (1998) document the diminishing returns to job changes over the course of the career, while Machado and Portela (2013) show that previous promotions are strong determinants of subsequent promotions. As such, the poor performance of visible minority immigrants in internal promotions may have consequences that cannot be offset by greater between-firm mobility over the long run. Indeed, Pendakur and Woodcock (2010) find that visible minority immigrants who have been in Canada for less than 10 years in the WES earn, on average, 31% less than similar white natives. While 42% of this wage gap is due to crowding of these immigrants into lower-paying firms as documented in Aydemir and Skuterud (2008), the remaining 58% is due to wage disparities relative to their native peers within firms. Our findings suggest that internal labor markets might play a key role in generating these within-firm wage gaps.

Second, why might visible minority immigrants be less "visible" to potential employers? The mobility patterns that we observe are consistent with potential employers having less information about visible minority immigrants. If information problems are at the heart of the mobility issues of visible minority immigrants, addressing this information asymmetry is important from a policy perspective. Given that the mobility difficulties appear to most pronounced among visible minority immigrants without higher education credentials, should policy-makers aim to vouch for the credentials (e.g., secondary school completion) of visible minority immigrants obtained abroad, or can credentialing programs be created? Can contacts with previous employers—potentially abroad—be facilitated?

Of course, information problems may be just one of many search frictions limiting the mobility of visible minority immigrants. Given the growing body of evidence—including our findings—of the existence of search frictions experienced by visible minority immigrants to Canada, understanding how their searches differ from those of their white peers (natives and immigrants) is of particular importance. Evidence concerning native-immigrant differences in Canada in the use of search networks, search methods, the geographic scope of search, and employer call-back rates would all shed much light on the potential existence and nature of the frictions experienced by visible minority immigrants in the Canadian labor market.

Endnotes

¹Studies documenting the occupational downgrading experienced by immigrants during their initial years in the host country include Chiswick (1978), Friedberg (2000), and Chiswick et al. (2005).

²Depew et al. (2017) study the between-firm mobility of skilled guest workers in the USA.

³Immigrants may lack the knowledge of local labor market institutions necessary for job search. Alternatively, immigrant enclaves might limit the search networks of recent immigrants to Canada (Warman 2007). If immigrants encounter search frictions not experienced by natives, employers may enjoy some degree of monopsony power over them that could drive the native-immigrants wage gap. Hirsch and Jahn (2015) and Naidu et al. (2016) provide evidence of employers' monopsony power over immigrants in Germany and the UAE, respectively.

⁴Other studies documenting the large early-career wage gains associated with job mobility include Bartel (1980), Borjas and Rosen (1980), Antel (1991), and McCue (1996).

⁵Imai et al. (2017), for instance, find that immigrants who arrived in Canada between 2000 and 2001 were initially employed in occupations requiring less cognitive skill and more manual skill than their occupations prior to immigration.

⁶The target population of employers consisted of all business locations in Canada with paid employees in March of each surveyed year. In the 1999, 2001, 2003, and 2005 surveys, the sample of employers was refreshed with new employers from the Statistics Canada Business Register to maintain a representative cross section. Employers in the Yukon, Nunavut, and Northwest Territories and employers operating in crop production, animal production, fishing, hunting, trapping, private households, religious organizations, and public administration were excluded from the sample. Public administration's share of employment in Canada is around 6.5% (Statistics Canada, Table 281-0024).

⁷The number of workers interviewed from each firm was proportional to firm's size except for workplaces with fewer than four employees in which all employees were surveyed.

⁸We identify promoted workers if they report having been promoted and the most recent promotion date is after the first interview.

⁹WES contains 47 detailed occupation categories based on the Standard Occupational Classification (SOC) 1991. Our occupational change indicator equals one if the detailed occupational category changes between interviews and zero otherwise.

¹⁰We focus on male workers because of differences in family formation between native and immigrant women. Javdani and McGee (2017) find that the promotion experiences of early-career women in the WES—particularly those with families—differ significantly from those of their male peers.

¹¹According to Statistics Canada (2011), the visible minority population in Canada consists mainly of Chinese, South Asian, Black, Arab, West Asian, Filipino, Southeast Asian, Latin American, Japanese, and Korean individuals. A worker is identified as a visible minority if her/his parents or grandparents belonged to one of these groups. The Employment Equity Act in Canada defines visible minorities to be "persons, other than Aboriginal people, who are non-Caucasian in race or non-white in color."

¹²Firms in the WES report the numbers of permanent full-time and part-time employees earning more than \$80,000, earning between \$60,000 and \$80,000, earning between \$40,000 and \$60,000, earning between \$20,000 and \$40,000, and earning less than \$20,000. We use this information along with the total number of employees within the firm to calculate the proportion of workers within the firm in a higher earnings category relative to any given worker. We cannot calculate the proportion of workers in higher earnings categories workers who earn more than \$80,000 (because no such category exists). For these workers, we set the proportion of workers in higher earnings categories to zero.

¹³Nearly 9% of white immigrants in the WES came from the USA compared to less than 1.5% of visible minority immigrants. Due to the North American Free Trade Agreement, workers from the USA need only a verifiable job offer from a Canadian employer to immigrate. The ease of return migration to the USA may explain the higher attrition rates of white immigrants.

¹⁴Kim (2012) develops sample and population attrition adjusted weights for application in short panels such as ours. He shows that the effect of population attrition in the CPS on assimilation estimates is minor. Given the similarity in his adjusted and unadjusted assimilation estimates and our short panel, we eschew the re-weighting procedure.

¹⁵The promotion rates in our sample are considerably higher than those reported in studies using changes in hierarchical levels or occupational categories to identify promotions (e.g., van der Klaauw and Dias da Silva 2011; Cassidy et al. 2016), but the promotion rates in those studies may fail to capture promotions within broad hierarchical levels or occupational categories. The promotion rates in our sample are similar to the rates of self-reported promotions among young workers in the USA documented in Pergamit and Veum (1999) and Cobb-Clark (2001).

¹⁶Consistent with their higher education levels, both white and visible minority immigrants earned more on average than white natives in our sample though this unconditional advantage is only statistically significant for white immigrants. Conditional on worker characteristics, visible minority immigrants earn significantly less than white native in terms of wage levels (see Appendix: Table 10).

¹⁷Unfortunately, our data do not identify to the immigration class to which an immigrant belonged.

¹⁸Many immigrants enter Canada on fixed length work permits prior to becoming permanent residents. The fixed duration may limit immigrants' promotion prospects if employers fear losing an employee when the permit expires, but work permits can be

renewed. In addition, there is no reason to expect that the effects of fixed term work permits on white and visible minority immigrants' promotion prospects would differ.

¹⁹This is unsurprising given that the predicted probabilities of the transition outcomes must sum to one for each group.

²⁰Between-firm mobility is likely influenced by local labor demand, and immigrants in Canada tend to be concentrated in provinces such as Ontario and British Columbia. As such, controlling for the region of residence is important in principle when estimating native-immigrant differences. Our early estimates, however, indicated that controlling for the province of residence and living in a city had no appreciable effect on the estimated marginal effects for immigrants.

²¹Oreopoulos (2011) indicates that recruiters rationalized their dismissal of the resumes of skilled immigrants based on language concerns.

²²Our choice of age 9 as the benchmark critical age is motivated in part by Corak's (2011) finding that children who immigrate to Canada after age 9 are much less likely to graduate from high school than those who immigrate at earlier ages.

²³The estimates reported in Table 3 come from a specification identical to that in column (2) of Table 2, but we replace the two immigrant indicators (white and visible minority) with four immigrant indicators (i.e., white and visible minority immigrants in the two age-at-immigration groups).

²⁴The shift to an immigration policy focused on admitting skilled immigrants was realized through several policy decisions in the early 1990s. We use 1993 as a benchmark because amendments to the Immigration Regulations in 1993 significantly reduced the share of family class immigrants.

²⁵Approximately 30% of the white immigrants in our sample come from the USA and the UK compared to only 3% of visible minority immigrants.

²⁶In terms of wage-level gaps, visible minority immigrants in the first interview earn more than white natives unconditionally but earn 15.1% less conditional on worker characteristics—a gap that grows to 16% by the second year. White immigrants, on the other hand, face no wage gaps relative to their Canadian-born counterparts in either year (see Appendix: Table 10).

²⁷See Javdani (2017) for a discussion of the low wage returns to promotion and low wage growth between interviews experienced by visible minority natives in the WES.

²⁸We use the procedure developed by Yun (2005) to transform the coefficients of the categorical transition dummies so that the results of the decomposition are invariant to the choice of the (omitted) base category. Alternative decomposition methods (e.g., using the coefficients from a pooled model over white natives and the minority group as the reference coefficients) produced similar results where the explained gaps were concerned.

²⁹One potential concern for our estimates is that immigrants and natives with the same amount of potential experience may have different amounts of Canadian labor market experience given that some immigrants come to Canada after their labor market entry. To assess the robustness of our findings, we reestimated the O-B decompositions restricting the sample to natives and immigrants who entered Canada within their first 3 years in the labor force (based on our potential experience measure). The estimates, reported in Appendix: Table 11, are similar to those in Table 7. Estimates from multinomial logit models of transition probabilities using this restricted sample are also similar to those reported in Table 2.

³⁰We also estimated probit models of the probability of occupation switching controlling for group indicators and worker characteristics. Similar to our multinomial logit estimates, the worker characteristics had little effect on the estimated marginal effects of the group indicators. As such, we report only the summary statistics by group in Table 8 for simplicity.

Appendix

	(1)	(2)	(3)	(4)	(5)
Visible minority Canadian-born	- 0.041	- 0.034	- 0.037	- 0.026	- 0.032
	(0.041)	(0.043)	(0.043)	(0.043)	(0.043)
White immigrant	0.076	0.125**	0.124**	0.132**	0.130***
	(0.049)	(0.052)	(0.050)	(0.052)	(0.050)
Visible minority immigrant	0.031	0.055	0.056	0.062	0.064
	(0.050)	(0.048)	(0.048)	(0.048)	(0.049)
Controls					
Person and job chars		Yes	Yes	Yes	Yes
Occupation			Yes		Yes
Industry				Yes	Yes

Table 9 Marginal effects from probit models of the probability of attrition

Notes: The sample includes observations from 6331 men interviewed in the first (odd) year with less than 10 years of potential experience. The table reports marginal effects from a probit model of the probability of attrition in which the dependent variable equals one if the respondent is interviewed in the odd year and not interviewed in the even year and zero otherwise

*Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

	(1)	(1)		(2)		(3)	
	Odd year	Even year	Odd year	Even year	Odd year	Even year	
Visible minority immigrants	0.054	0.036	- 0.151***	- 0.160***	- 0.127***	- 0.140***	
	(0.053)	(0.055)	(0.051)	(0.054)	(0.044)	(0.040)	
White immigrants	0.219***	0.231***	0.041	0.041	0.017	0.020	
	(0.073)	(0.065)	(0.047)	(0.045)	(0.042)	(0.041)	
Visible minority Canadian-born	0.116	0.042	0.143***	0.055	0.121**	0.016	
	(0.084)	(0.085)	(0.047)	(0.044)	(0.048)	(0.042)	
Controls							
Worker characteristics			Y	es	Y	es	
Occupation and industry					Y	es	

Table 10 Wage gaps relative to white Canadian-born workers among men with less than 10 years of potential experience

Notes: Robust standard errors accounting for clustering at the firm level are reported in parentheses. The sample is restricted to the 4585 men with valid wage observations in both interviews

*Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

	Visible minority immigrants	White immigrants	Visible minority Canadian-born
	(1)	(2)	(3)
Log wage growth for white Canadians	0.083***	0.083***	0.083***
	(0.007)	(0.007)	(0.007)
Log wage growth for minority group	0.052*	0.089***	0.009
	(0.029)	(0.030)	(0.030)
Raw gap in log wage growth	0.030	- 0.005	0.073**
	(0.030)	(0.031)	(0.031)
Explained	- 0.008	- 0.024*	- 0.004
	(0.010)	(0.014)	(0.011)
Unexplained	0.039	0.018	0.078**
	(0.029)	(0.027)	(0.032)
Explained by differences in characteristics			
Transition: new employer	- 0.002	- 0.012	0.002
	(0.009)	(0.011)	(0.006)
Transition: promoted	0.004*	0.001	0.002
	(0.002)	(0.001)	(0.002)
Demographics	- 0.001	- 0.005	0.007*
	(0.008)	(0.007)	(0.004)
Education	- 0.006	- 0.003	- 0.005
	(0.007)	(0.005)	(0.004)
Experience	- 0.002	- 0.002	0.003
	(0.002)	(0.002)	(0.004)
Occupation	0.005	0.007	0.000
	(0.005)	(0.005)	(0.005)
Industry	- 0.007	- 0.004	- 0.010
	(0.004)	(0.004)	(0.006)
Explained by differences in returns to cha	aracteristics		
Transition: new employer	0.009	0.006	0.013
	(0.016)	(0.015)	(0.016)
Transition: promoted	0.000	0.008	0.025
	(0.013)	(0.011)	(0.022)
Demographics	0.124	- 2.539***	1.146
	(1.342)	(0.948)	(1.779)
Education	- 0.002	- 0.006	- 0.020
	(0.044)	(0.019)	(0.040)
Experience	0.029	0.024	0.015
	(0.088)	(0.072)	(0.060)
Occupation	- 0.008	- 0.009	0.016
	(0.027)	(0.024)	(0.022)
Industry	0.014	0.000	- 0.067*
	(0.034)	(0.025)	(0.036)

Table 11 Robustness of O-B decompositions

Notes: The sample includes only immigrants who entered Canada prior to their first 3 years in the labor force. See the notes to Table 7 for additional details

*Significant at the 10% level; **significant at the 5% level; ***significant at the 1% level

Acknowledgements

We thank the participants at the annual meetings of the Canadian Economics Association, the Western Economics Association, and the Society of Labor Economics for their suggestions. We would also like to thank the anonymous referee and the editor for the useful remarks. All remaining errors are our own. Responsible editor: Hartmut F. Lehmann

Availability of data and materials

The data used in our study are confidential data that were made available through the Statistics Canada Research Data Centre program. RDCs provide researchers with access, in a secure university setting, to microdata from population and household surveys. The centers are staffed by Statistics Canada employees. They are operated under the provisions of the Statistics Act in accordance with all the confidentiality rules and are accessible only to researchers with approved projects who have been sworn in under the Statistics Act as "deemed employees." Application process and guidelines to get access to the data are based on the affiliation of the Principal Investigator and the type of research being conducted at a Research Data Centre (RDC) and are available here: http:// www.statcan.gc.ar/dc-cdr/process-eng.htm.

Competing interests

The IZA Journal of Development and Migration is committed to the IZA Guiding Principles of Research Integrity. The authors declare that they have observed these principles.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Department of Economics, 3333 University Way, Kelowna, BC V1V 1V7, Canada. ²Department of Economics, University of Alberta, Tory Building, Edmonton, AB T6G 2H4, Canada. ³IZA - Institute of Labor Economics, Bonn, Germany.

Received: 17 November 2017 Accepted: 3 April 2018 Published online: 25 October 2018

References

- Abramitzky R, Boustan L, Eriksson K. A nation of immigrants: assimilation and economic outcomes in the age of mass migration. J Polit Econ. 2014;122(3):467–506.
- Adsera, Alicia, and Ana M. Ferrer. 2015. The effect of linguistic proximity on the occupational assimilation of immigrant men in Canada. Working paper.
- Antel JJ. The wage effects of voluntary labor mobility with and without intervening unemployment. Ind Labor Relat Rev. 1991;44:299–306.
- Antonovics K, Golan L. Experimentation and job choice. J Labor Econ. 2012;30:333-66.

Aydemir A, Skuterud M. The immigrant wage differential within and across establishments. Ind Labor Relat Rev. 2008; 61(3):334–52.

Bartel AP. Earnings growth on the job and between jobs. Econ Ing. 1980;18(1):123-37.

Blau FD, DeVaro J. New evidence on gender differences in promotion rates: an empirical analysis of a sample of new hires. Ind Relat J Econ Soc. 2007;46(3):511–50.

- Bleakley H, Chin A. Language skills and earnings: evidence from childhood immigrants. Rev Econ Stat. 2004; 86(2):481–96.
- Blinder AS. Wage discrimination: reduced form and structural estimates. J Hum Resour. 1973;8(4):436–55.

Borjas GJ, Rosen S. Income prospects and job mobility of younger men. Res Labor Econ. 1980;3:159–81.

- Bowlus AJ, Miyairi M, Robinson C. Immigrant job search assimilation in Canada. Can J Econ. 2016;49(1):5-51.
- Burdett K. A theory of employee job search and quit rates. Am Econ Rev. 1978;68(1):212–20.
- Cassidy H, DeVaro J, Kauhanen A. Promotion signaling, gender, and turnover: new theory and evidence. J Econ Behav Organ. 2016;126:140–66.

Chiswick B. The effect of Americanization on the earnings of foreign-born men. J Polit Econ. 1978;86:897–922.

- Chiswick BR, Lee YL, Miller PW. A longitudinal analysis of immigrant occupational mobility: a test of the immigrant assimilation hypothesis. Int Migr Rev. 2005;39(2):332–53.
- Chiswick BR, Miller PW. The endogeneity between language and earnings: international analyses. J Labor Econ. 1995; 13(2):246–88.
- Chui T. Immigration and ethnocultural diversity in Canada: National Household Survey, 2011. Ottawa: Statistics Canada; 2013.
- Citizenship and Immigration Canada. 2011. Evaluation of the provincial nominee program. Evaluation and research. Accessed January 12, 2018.
- Cobb-Clark DA. Getting ahead: the determinants of and payoffs to internal promotions for young US men and women. Research in Labor Economics. 2001;20:339–72.
- Corak, Miles. 2011. Age at immigration and the education outcomes of children. IZA discussion paper 6072. Depew B, Norlander P, Sorensen TA. Inter-firm mobility and return migration patterns of skilled guest workers. J Popul
- Econ. 2017;30(2):681–721. Dustmann C, van Soest A. Language and the earnings of immigrants. Ind Labor Relat Rev. 2002;55(3):473–92.

Francesconi M. Determinants and consequences of promotions in Britain. Oxf Bull Econ Stat. 2001;63(3):279-310.

Friedberg RM. You can't take it with you? Immigrant assimilation and the portability of human capital. J Labor Econ. 2000;18(2):221–51.

Green AG, Green DA. The economic goals of Canada's immigration policy: past and present. Can Public Policy. 1999; 25(4):425–51.

- Green D. Immigrant occupational attainment: assimilation and mobility over time. J Labor Econ. 1999;17(1):49-79.
- Green D, Worswick C. Immigrant earnings profiles in the presence of human capital investment: measuring cohort and macro effects. Labour Econ. 2012;19(2):241–59.
- Grenier G, Xue L. Canadian immigrants' access to a first job in their intended occupation. J Int Migr Integr. 2011;12(3): 275–303.
- Hirsch B, Jahn EJ. Is there monopsonistic discrimination against immigrants? Ind Labor Relat Rev. 2015;68(3):501-28.

Imai S, Stacey D, Warman C. From engineer to taxi driver? Occupational skills and the economic outcomes of immigrants. Can J Econ. 2017; in press

- Javdani, Mohsen. 2017. Does color matter? Estimating differences in promotions and returns to promotions between white and visible minority Canadian-Borns. Working paper.
- Javdani, Mohsen and Andrew McGee. 2017. Moving up or falling behind? Gender, promotions, and wages in Canada. Working paper.
- Jovanovic B. Job matching and the theory of turnover. J Polit Econ. 1979;87:972–90.

Kim, Seik. 2012. Sample attrition in the presence of population attrition. Working paper.

Kosteas VD. Job level changes and wage growth. Int J Manpow. 2009;30(3):269-84.

- Krebs H, Patak Z, Picot G, Wannell T. The development and use of a Canadian linked employer-employee survey. In: The creation and analysis of employer-employee matched data. Ottawa: Statistics Canada; 1999. p. 515–34.
- Light A, McGarry K. Job change patterns and the wages of young men. Rev Econ Stat. 1998;80(2):276–86.

Machado, C. Sofia, and Miguel Portela. 2013. Age and opportunities for promotion IZA discussion paper 7784 McCue K. Promotions and wage growth. J Labor Econ. 1996;14(2):175–209.

- Milgrom P, Oster S. Job discrimination, market forces, and the invisibility hypothesis. Q J Econ. 1987;102(3):453–76.
 Naidu S, Nyarko Y, Wang S-Y. Monopsony power in migrant labor markets: evidence from the United Arab Emirates. J Polit Econ. 2016;124(6):1735–92.
- Oaxaca R. Male-female wage differentials in urban labor markets. Int Econ Rev. 1973;14(3):693-709.
- Oaxaca R, Ransom M. Identification in detailed wage decompositions. Rev Econ Stat. 1999;81(1):154-7.
- Oreopoulos P. Why do skilled immigrants struggle in the labor market? A field experiment with six thousand résumés. Am Econ J Econ Pol. 2011;3(4):148–71.
- Pendakur K, Woodcock S. Glass ceilings or glass doors? Wage disparity within and between firms. J Bus Econ Stat. 2010; 28(1):181–9.

Pergamit MR, Veum JR. What is a promotion? Ind Labor Relat Rev. 1999;52(4):581-601.

Singleton D, Lengyel Z, editors. The age factor in second language acquisition: a critical look at the critical period hypothesis. Philadelphia: Clevedon [England]; 1995. p. 124–46.

Skuterud M, Su M. Immigrants and the dynamics of high-wage jobs. Ind Labor Relat Rev. 2012;65(2):377–97.

- Statistics Canada. 2011. Visible minority and population group reference guide, National Household Survey.
- Statistics Canada. 2016. Table 051-0011—international migrants, by age group and sex, Canada, provinces, and territories, annual (persons), CANSIM (database). (accessed: April 10, 2017).
- Topel RH, Ward MP. Job mobility and the careers of young men. Q J Econ. 1992;107(2):439-79.
- van der Klaauw B, Dias da Silva A. Wage dynamics and promotions inside and between firms. J Popul Econ. 2011;24: 1513–48.
- Wald S, Fang T. Overeducated immigrants in the Canadian labour market: evidence from the workplace and employee survey. Can Public Policy. 2008;34(4):457–79.

Warman C. Ethnic enclaves and immigrant earnings growth. Can J Econ. 2007;40(2):401-22.

Yun M-S. A simple solution to the identification problem in detailed wage decompositions. Econ Inq. 2005; 43(4):766–72.

Submit your manuscript to a SpringerOpen[®] journal and benefit from:

- Convenient online submission
- ► Rigorous peer review
- Open access: articles freely available online
- ► High visibility within the field
- Retaining the copyright to your article

Submit your next manuscript at > springeropen.com